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19 April 2010
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 16 of 2010, which covers the period from 11 through 17 April 2010.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 109.9 to 123.2 kcfs.  Daily average project tailwater ranged from 10.8’ to 11.5’ msl.  Daily average powerhouse forebay elevation ranged from 75.8’ to 76.4’ msl.  Daily average river temperature ranged from 47 to 49ºF.  Secchi disk measurements ranged from 4.0’ to 6.0’.
1.2 Spill Bays 1 and 18, which provide attraction flow for Cascades Island and B-branch, respectively, are open 6” during daylight hours.  Daily average spill ranged from 67.0 to 80.2 kcfs.
1.3 Unit Operation.  Powerhouse Two is the priority powerhouse. 
1.3.1 Unit 5 was out of service from 1256 until 1709 on 12 April due to replacing electric turbine bearing pump, and from 0800 until 1553 on 14 April due to turbine bearing oil pump repair.
1.3.2 Unit 7 has remained out of service since 04 June 2007 for turbine rehab.

1.3.3 Unit 9 has remained out of service since 04 August 2008 for turbine rehab.
1.3.4 Unit 11 has remained out of service since 12 June 2008 due to rotor damage. 

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore (WS) fish ladder.  The units may be placed on standby, as necessary, during nighttime hours to float debris.
1.5 Gatewell drawdowns are measured on Wednesdays.
2 ACTIVITIES

2.1 Auxiliary Water System Closures.  The following AWS valves were closed this reporting week:
2.1.1 FV 3-9 was closed for cleaning from 1458 until 1505 on 15 April.
2.1.2 FV 5-9 and FV 6-9 were closed for cleaning from 1524 until 1541 on 15 April.
2.2 STS/VBS inspections.  No STS/VBS inspections occurred this reporting week.
Table 1.  STS and VBS inspections.  
	Unit
	Date
	Run Hours
	Comments

	11
	NA
	OOS
	U11 remains out of service.

	13-17
	NA
	NA
	Inspection scheduled for week of 26 April

	12 & 18
	4/12
	NA
	VBSs cleaned.


2.3 Dewatering and Fish Salvage.  No dewatering/fish salvage activities occurred this week.
3 RESEARCH  

3.1 Battelle – survival study.  Hydrophone installation in PH1, PH2, and spillway forebays continues.

3.2 CRITFC- sea lion. Deployed hydrophones on 01 March in BON tailraces to track sea lion activity.

3.3 CRITFC - Annual chinook, coho, sockeye and steelhead sampling in the AFF began this week.

3.4 FFU- sea lion.  Observations continue.  Weekly reports of sea lion activities are now available. If you would like to be included on the distribution list contact Robert Stansell-541-374-8801. 

3.5 NOAA- lamprey.  Cascades Island lamprey ramp camera remains installed.
3.6 PSMFC- smolt monitoring.  Juvenile fish condition and index monitoring continues.  Fallbacks recorded crossing the separator bars for this reporting week:  six steelhead.  Please see http://www.fpc.org/adultsalmon/bonkeltcounts.htm for updated kelt counts.
3.7 Smith-Root- sea lion array.  Array remains installed in the UMT.  Telemetry receivers and DIDSON remain installed.  U of I is scheduled to test affects of array on radio tags the week of 19 April.
3.8 WDFW- Fish Counting.  Visual counting continues.
3.9  WDFW, ODFW, CRITFC- Sea lion hazing/trapping.  Hazing from boats continues.  Sea lion traps remain in position.  Trapping continues.
3.10  USDA Wildlife Services- Sea lion hazing.  Hazing from shore continues.

3.11 USDA Wildlife Services- Avian hazing.  Hazing from shore began on 01 April.

3.12 University of Idaho – lamprey.   No juvenile lamprey collected this reporting week. 

3.13 University of Idaho - Adult salmon sampling continues for monitoring passage effects at the modified Cascades Island entrance.
4 FISHWAYS 

4.1 Project biologists inspected 11-17 April.
4.2  Fish Passage Plan Observations.
Table 2.  FPP Out of Criteria items.
	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND (BI)

	4/12,16
	A-branch weir
	Staff gage > 1.0’
	Unknown
	No action.
	

	CASCADES ISLAND (CI) - Nothing to report

	WASHINGTON SHORE (WS) – Nothing to report


4.2.1 PH1 PCC diffuser FG2-19 is partially open.  Repairs scheduled for 2011/2012 in water work period.  

4.2.2 Design of a mechanical velocity meter at the PH2 south monolith is in progress.  

4.3 Fishway Temperatures.  Temperatures are taken 01 March through 30 November.

Table 3.  Bonneville Fishway Temperatures

	Date
	WA. Shore average
	BI

average

	4/11
	46.1
	46.9

	4/12
	46.6
	47.6

	4/13
	46.9
	47.9

	4/14
	47.1
	47.9

	4/15
	47.6
	48.4

	4/16
	48.2
	49.0

	4/17
	48.4
	49.4


4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  
4.5 Adult Fishways. 
4.5.1 Calibration was checked on 16 April.
4.5.2 The Washington Shore fishway entrances remain in manual due to SA-24 board calibration issues.  The operators are adjusting the gates to maintain at least a 13’ opening, as tailwater allows.
4.5.3 Sea lion exclusion bars remain in 11 of 12 FOGs.  One gate remains out for maintenance, with a bulkhead installed.
4.5.4 Sea Lion Exclusion Devices (SLEDs) remain installed at all fishway entrances. 
4.5.5 Sampling at the AFF continues.
4.6 Juvenile Fishways.
4.6.1 The B2 Corner Collector remains open. 
4.6.2 DSM2 remains in service.
4.6.3 The PH1 ice and trash sluiceway (ITS) remains in service.  
4.6.4 Hydrocannon remains in service.
4.6.5 TIES are installed.
4.7 Lamprey Fishways.  All LPSs are dewatered.
4.7.1 The Cascades Is. entrance weir remains installed.  
4.7.2 The PH2 NDE ramp remains in the lowered position.
5 CONSTRUCTION
5.1 PH1 Gantry 4.  New crane in FY11.  
5.2 PH1 Ice and Trash Sluiceway Wall Removal. Wall removal is complete.  Outfall removal has been postponed.
5.3 Spillway Gate Repair Pit. Clean-up of hazardous material is in progress.  Structural modifications scheduled to occur in late summer.
6 HAZMAT, SPILLS AND CLEANUP.  Nothing to report.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A teletype sent on 16 July 2009, changes PH2 unit priority to: 12,18,15,17,14,13,16.  This change is effective from 1100 on 16 July 2009 until U11 returns to service.
7.2 A teletype sent on 09 April requests spill for juvenile fish passage begin at 0001 hours on Saturday, 10 April and continue until further notice. Spill approximately 100 kcfs 24 hours a day using the spill pattern in the 2010 FPP.  If Bonneville inflows are too low to maintain the 100 kcfs spill and minimum turbine discharge of 30 kcfs, then operate to minimum discharge and spill to pass inflow.  

7.3 A teletype sent on 09 April requests special operations for Spring Creek and Little White Salmon hatchery releases on 12 April and 13 April respectively.  Beginning 0600 on 13 April until 0559 on 17 April as a hard constraint and from 0600 on 17 April until 0600 on 21 April as a soft constraint, operate PH2 units at the low 1% best efficiency range.  

7.4 A teletype sent on 16 April requests that the B2CC close from 0630 until 1659 on 20 April.  If this operation is unable to occur on 20 April, then this same operation can be done on 21 April.  This operation is to support the installation of research equipment in the PH2 forebay, and repair the damaged float cable on the BGS.
7.5 A teletype sent on 16 April modifies the teletype sent on 09 April (see 7.3).  Beginning at 0600 on 13 April until 1659 on 19 April as a hard constraint, and from 1700 on 19 April until 1700 on 21 April as a soft constraint, operate PH2 units at the low 1% best efficiency range.  This operation was requested for Spring Creek and Little White Salmon hatchery fish releases on 12 and 13 April.
GLEN A. SMITH, P.E.
Operations Project Manager
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